Author's response to reviews

Title: Excessive Use of dangerous forms of Alternative medicine by the Saudi patients with neurological disorders: skin cauterization and bloodletting

Authors:

Yousef M Mohammad (ymohammad@ksu.edu.sa)
Ahmed N Al-Ahmari (An1410@gmail.com)
Fahad A Al-Dashash (F.a.dashash@gmail.com)
Fawaz A Al-Hussain (Faalhussain@ksu.edu.sa)
Firas F Al-Mansour (Firas_dm@hotmail.com)
Abdullah A Masoud (A_stonecold_a@hotmail.com)
Hoda a Jradi (jradiho@NGHA.MED.SA)

Version: 3 Date: 17 February 2014

Author's response to reviews:

The consent to publish the pictures in the two figures were obtained from the patients. The consent form, including the patients' names and signatures, is uploaded.

The authors' contribution is now included in the manuscript, instead of the cover letter, as you suggested.

The authors declare that they have no competing interests.

A translated copy of the questionnaire into English is uploaded.

Thank you,

Yousef Mohammad MD., MSc.